Author's response to reviews

Title: Verification of genes differentially expressed in neuroblastoma tumours: a study of potential tumour suppressor genes.

Authors:

Kaisa Thorell (kaisa.thorell@wlab.gu.se)
Annika Bergman (annika.bergaman@clingen.gu.se)
Helena Carén (helena.caren@clingen.gu.se)
Staffan Nilsson (staffan@chalmers.se)
Per Kogner (Per.Kogner@ki.se)
Tommy Martinsson (tommy.martinsson@clingen.gu.se)
Frida Abel (frida.abel@clingen.gu.se)

Version: 4 Date: 7 August 2009

Author's response to reviews:

Gothenburg, Sweden
August 7th, 2009

Dear Editor,

We are pleased to hear that our manuscript entitled: “Verification of genes differentially expressed in neuroblastoma tumours: a study of potential tumour suppressor genes” has been accepted for publication in BMC Medical Genomics.

We have now made all formatting changes as requested.

On behalf of all authors,

Yours sincerely,

Frida Abel
Ph.D.